Author's response to reviews

Title: Urine alkalization facilitates uric acid excretion.

Authors:

Aya Kanbara (achan@gaines.hju.ac.jp)
Masayuki Hakoda (hakoda@yasuda-u.ac.jp)
Issei Seyama (seyamai@gaines.hju.ac.jp)

Version: 5 Date: 11 October 2010

Author's response to reviews: see over
Dear Sir:

We corrected all the places you pointed out.

Corrected places are listed as follows,

1. removed full stop from the title.
2. removed ‘by’ from the author list.
3. erased all red underlines.
4. removed blank page.
5. added the word ‘being’ in the line 5 from the bottom on page 7 in the statistical analysis section.
6. added the letter ‘s’ to ‘asterisk’ for making it plural.
7. changed all capital letters to small letters in the appendix section.

With my best regards.